

Citation: Bogoch II, Speich B, Lo NC, Moser W, Croll D, Ali SM, et al. (2019) Clinical evaluation for morbidity associated with soil-transmitted helminth infection in school-age children on Pemba Island, Tanzania. PLoS Negl Trop Dis 13(7): e0007581. https://doi.org/10.1371/journal.pntd.0007581

Editor: Jennifer A. Downs, Weill Cornell Medical College, UNITED STATES

Received: February 7, 2019

Accepted: June 26, 2019

Published: July 15, 2019

Copyright: © 2019 Bogoch et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information file.

Funding: IIB was supported by a grant from the MSH UHN AMO Innovation Fund. The clinical trial was supported by the Medicor Foundation and the Swiss National Science Foundation (SNSF; Bern, Switzerland; grant number: 320030_14930). The funders played no role in the study design, data collection and analysis, decision to publish, or preparation of the manuscript. RESEARCH ARTICLE

Clinical evaluation for morbidity associated with soil-transmitted helminth infection in school-age children on Pemba Island, Tanzania

Isaac I. Bogoch ^{1,2}*, Benjamin Speich ^{3,4,5}, Nathan C. Lo^{6,7}, Wendelin Moser^{3,4}, David Croll^{3,4}, Said M. Ali⁸, Shaali M. Ame⁸, Jürg Utzinger^{3,4}, Jason R. Andrews⁷, Jennifer Keiser^{3,4}

 Divisions of General Internal Medicine and Infectious Diseases, Toronto General Hospital, University Health Network, Toronto, Canada, 2 Department of Medicine, University of Toronto, Toronto, Canada,
Swiss Tropical and Public Health Institute, Basel, Switzerland, 4 University of Basel, Basel, Switzerland,
Basel Institute for Clinical Epidemiology and Biostatistics, Department of Clinical Research, University of Basel and University Hospital Basel, Basel, Switzerland, 6 Department of Medicine, University of California, San Francisco, San Francisco, California, United States of America, 7 Division of Infectious Diseases and Geographic Medicine, Stanford University School of Medicine, Stanford, California, United States of America, 8 Laboratory Division, Public Health Laboratory - Ivo de Carneri, Chake Chake, Tanzania

* isaac.bogoch@uhn.ca

Abstract

Background

More than 1.5 billion people are infected with soil-transmitted helminths (*Ascaris lumbri-coides*, hookworm, *Strongyloides stercoralis*, and *Trichuris trichiura*), causing an estimated global burden in excess of 3 million disability-adjusted life years. However, the relationship between soil-transmitted helminth infection, adverse health consequences, and beneficial effects of deworming are not well understood.

Methodology

We pursued a detailed longitudinal clinical evaluation of school-age children to evaluate morbidity associated with soil-transmitted helminth infection and responses to treatment. This exploratory study was embedded into a randomized controlled trial. Overall, 434 children, aged 7–14 years, underwent a detailed medical history, physical examination, stool microscopy for soil-transmitted helminths, and hemoglobin (Hb) measurement at baseline. Medical history and stool examination were repeated at 3 and 18 weeks posttreatment. Additionally, Hb measurement was performed at the 18-week treatment follow-up. Logistic regression was employed to assess clinical factors associated with soil-transmitted helminth infection at baseline, and longitudinal data analysis to examine change in health outcomes following treatment over time.

Principal findings

All enrolled children were infected with *T. trichiura*, and randomized into four different treatment interventions. None of the medical history, physical examination, and laboratory (i.e., PLOS | NEGLECTED TROPICAL DISEASES

Competing interests: The authors have declared that no competing interests exist.

Hb) findings were associated with *A. lumbricoides*, hookworm, or *S. stercoralis* infection at baseline. A composite of physical exam findings for anemia, including pallor of the conjunctiva, nail beds, and palmar creases predicted lower Hb values (-3.8 g/dl, 95% confidence interval (CI): -6.9, -0.6 g/dl). When examining longitudinal trends, we did not find improvements to Hb or face Wong-Baker Likert scale among children with soil-transmitted helminth infection compared to those without infection, although there was a slight trend toward improving Hb values after treating hookworm infection.

Conclusions/significance

Our study demonstrates the challenges of measuring morbidity in the context of soil-transmitted helminth infection and treatment, thus confirming the mainly subtle morbidity effects of infection.

Author summary

Soil-transmitted helminth infections frequently affect children and impoverished individuals in low-resource settings and are associated with negative health effects. We attempted to measure morbidity in school-age children from Pemba Island, Tanzania who are infected with soil-transmitted helminths but with relatively low risk for other infections known to cause chronic morbidity, such as malaria and schistosomiasis. We used a composite of clinical history, physical examination, and hemoglobin measurements. Our study was embedded into a randomized controlled trial, and by design, all individuals were infected with whipworms (*Trichuris trichiura*). A detailed clinical evaluation for morbidity in school-age children was not associated with an infection of roundworms (*Ascaris lumbricoides*), hookworm, or threadworms (*Strongyloides stercoralis*) but demonstrated the utility of the clinical examination to detect anemia. Indeed, anemia slightly trended toward resolution after deworming. While many metrics for soil-transmitted helminth infection-related morbidity may have been masked by ongoing deworming campaigns, simple physical exam findings for anemia can be useful to highlight individuals that require further evaluation.

Introduction

Soil-transmitted helminths (*Ascaris lumbricoides*, hookworm, *Strongyloides stercoralis*, and *Trichuris trichiura*) affect more than 1.5 billion people worldwide with the highest prevalence and intensity of infection often observed in school-age children in low- and middle-income countries [1, 2]. Chronic infection with soil-transmitted helminths can lead to anemia, malnutrition, stunted growth, and delayed cognitive development [2]. The current global strategy to control infection in high-burden areas aims to reduce morbidity with preventive chemotherapy in regions where the prevalence of infection is above a pre-specified threshold [3, 4], however, re-infection rates remain high [5, 6]. There is some debate over the role of preventive chemotherapy in reducing morbidity in school-age children, and hence, there is a need for rigorous studies to evaluate the impact of soil-transmitted helminths on morbidity and changes after treatment [7, 8]. It is still generally accepted that preventive chemotherapy is beneficial and that contradictory results may be related to how and when morbidity is measured [9–11]

but a better clinical understanding of measurable reductions in soil-transmitted helminth infection-related morbidity in response to therapy is now recognized as a priority research area [12].

Several non-invasive strategies have been evaluated to either diagnose soil-transmitted helminth infection or determine the morbidity associated with infection, including the use of anamnestic questionnaires [13] and ultrasonography [14]. To date, studies assessing the utility of a clinical examination aimed at diagnosis or quantifying morbidity associated with soil-transmitted helminth infection have been somewhat limited in scope. Here, we report on the results of a detailed clinical evaluation of school-age children infected with one or several species of soiltransmitted helminths in rural communities on Pemba Island, Tanzania. We evaluated whether a clinical approach was able to detect morbidity associated with soil-transmitted helminth infection, and whether changes to morbidity are clinically detectable in response to treatment.

Methods

Ethics statement

Ethical approval for the study was granted by both the *Zanzibar* Medical Ethical Research Committee (reference no. ZAMREC, 0001) and the Ethics Committee of Northwestern and Central Switzerland (reference no. EKBB #123). Children aged 7–14 years from two schools (Mchangamdogo and Shungi) were invited to enroll in this study. All participants had written informed consent from a parent or legal guardian, while children assented orally. Participation was voluntary and children could withdraw anytime without further obligations.

Study site and design

The study was integrated into a randomized controlled trial (RCT), evaluating novel treatment strategies for *T. trichiura* infection in school-age children on Pemba Island, Tanzania. It was carried out between September 2013 and March 2014. Details of the design of the RCT, base-line characteristics, and infection data of school-age children, and results of the efficacy and safety of the different treatment strategies have been reported elsewhere [15, 16].

Selection for the RCT started before this study, and all children enrolled were, by design, infected with *T. trichiura* at baseline. The study consisted of a baseline and two follow-up assessments at 3 and 18 weeks after randomization and initial anthelmintic treatment. The baseline evaluation included a detailed medical history, physical examination, stool microscopy for diagnosis of species-specific helminth infection and intensity, and hemoglobin (Hb) finger prick measurement with a HemoCue device (HemoCue 301 system; Ängelholm, Sweden). Children were randomized and treated with either mebendazole, albendazole plus mebendazole, albendazole plus ivermectin, or albendazole plus oxantel pamoate. The first follow-up was performed 3 weeks after randomization and treatment, and included a detailed medical history and stool microscopy. A second follow-up was conducted 18 weeks after baseline testing, and included a detailed medical history, stool microscopy, and a repeat Hb measurement. Hb measurements were conducted at this time to allow for an appropriate time to measure the potential resolution of anemia, as assessed by Hb level [17]. At the end of our study, after the 18-week follow-up survey, all children, regardless of their infection status, were treated with albendazole (400 mg) as per national guidelines.

Clinical evaluation

A dedicated school classroom was temporarily converted into a clinic space, where clinical evaluations were conducted, and to facilitate laboratory specimen collection. Trained nurses

using a standardized questionnaire conducted the medical history component of the clinical examination, directly in Swahili [18]. The first section of the medical history highlighted past medical history, allergies, medication history, and any illness in the 4 weeks before the survey that were severe enough for a child to stay home from school. The medical history then focused on generalized and focal issues, including active symptoms of fever, chills, cough, headache, vertigo, abdominal cramps, fatigue, nausea, vomiting, diarrhea, hematochezia, hematuria, constipation, anorexia, pruritis, ankle edema, dyspnea, and difficulty concentrating. If there were active symptoms present, children were asked to elaborate on the severity and duration of such symptoms. Lastly, children were asked to point to a Wong-Baker Likert scale of six faces [19, 20], that best described their overall health and wellbeing over the past 4 weeks.

Two experienced physicians performed a detailed physical examination. This focused on general characteristics (height, weight, heart rate, respiratory rate, and presence of jaundice), and appearance of malnourishment (temporalis muscle wasting, loss of subcutaneous fat at the deltoid, triceps, interosseous hand muscles, and quadriceps muscle groups [21], and conjunctival evidence of vitamin A deficiency). Children were then evaluated for the presence of abdominal pain with light and deep palpation in four quadrants, the presence of ascites (inspection, pitting ankle edema, flank dullness, and fluid wave [22]), splenomegaly (inspection, Castell's sign, and palpation [23]), hepatomegaly (inspection, palpation, and hook test [24]), pulmonary hypertension (cyanosis, prominent "a" or "cv" waves on jugulovenous pressure, Kussmaul's sign, palpable P2 on precordial exam, a murmur of tricuspid regurgitation, or the presence of right sided S3 or S4 heart sound [25]) in addition to a general cardiac and pulmonary exam via auscultation. Auscultation was performed with double lumen stethoscopes (Littmann, 3M; St. Paul, MN, United States of America). Lastly, signs of anemia were recorded and included evaluation for pallor of the conjunctiva, nail beds, and palmar creases, in addition to evidence of koilonychia, angular chelitis, or glossitis [26, 27]. Data were collected on paper forms and then entered in duplicate into Microsoft Excel files (Microsoft; Redmond, WA, United States of America).

Laboratory procedures

Children provided stool samples on two consecutive days for baseline, first follow-up at 3 weeks after randomization and treatment, and second follow-up at 18 weeks posttreatment. Samples were transferred to the Public Health Laboratory—Ivo de Carneri for processing on the day of collection. Duplicate Kato-Katz thick smears were made from each stool sample on both days, using a 41.7 mg template [28]. Kato-Katz thick smears were examined under a microscope by one of six experienced laboratory technicians for the presence and quantification of eggs from *A. lumbricoides*, hookworm, and *T. trichiura*. All stool samples were processed the same day and slides read under a microscope within one hour of preparation to account for rapid hookworm egg disintegration [29]. For quality control, 10% of slides were randomly selected and re-read by a third, senior microscopist and the results were discussed until consensus was reached. Additionally, stool samples were examined for *S. stercoralis* infection via Koga-agar plate and Baermann techniques [30]. The presence of larvae by one or both of these techniques constituted a positive test.

Statistical analysis

We included children who had written informed consent from their parents/legal guardians, assented orally, and had complete data records for their parasitologic (quadruplicate Kato-Katz thick smear readings for all three time points), and clinical history and physical exam

outcomes. We reported helminth-specific prevalence and arithmetic mean and their corresponding 95% confidence intervals (CIs) for infection intensity (as expressed by eggs per 1 g of stool; EPG) at the three study time points. Using data on clinical history and physical exam, we assessed clinical factors associated with each soil-transmitted helminth infection. We selected clinical variables that were pre-specified based upon clinical expertise and excluded rarely reported outcomes (<2% of study population). We used six pre-defined aspects of the history based on clinical opinion (abdominal pain, blood in stool, diarrhea, headache, recent sick days, and Wong-Baker Likert scale), three physical exam features (a composite of anemia findings and malnutrition findings, and height), and one laboratory parameter (Hb) to determine clinical associations for infection with A. lumbricoides, hookworm, or S. stercoralis. Clinical variables associated with T. trichiura could not be conducted as, by design of the RCT, only children with an infection of *T. trichiura* at baseline were included. Examination for abdominal pain on physical examination was not used as all evaluations were benign. There were a total of 30 statistical comparisons (10 history, physical exam, and laboratory factors-including composite physical exam findings, with three binary parasitologic infection outcomes with each helminth species) in univariate analyses with logistic regression where infection with each soil-transmitted helminth was the dependent variable. We applied a Bonferroni correction to adjust for multiple comparisons for interpretation of our conclusions.

We investigated the longitudinal change in pre-defined health outcomes (Hb and face Wong-Baker Likert scale) following treatment using the two follow-up time points. For Hb, where only two measurements were available (baseline and 18-week posttreatement followup), we used an unpaired *t*-test to examine change in Hb in individuals with and without each helminth infection at baseline. We also examined the relation between change in Hb and change in infection intensity, as expressed in EPG with the Spearman correlation coefficient given overdispersion of helminth egg output values. For Wong-Baker Likert scale, where three measurements were available, we used a generalized estimating equation with exchangeable correlation structure and robust standard errors to examine the longitudinal relationship between changes in Wong-Baker Likert scale (dependent variable) over time and baseline helminth infection status (independent variable). Additionally, Wong-Baker Likert scores and Hb levels for each treatment arm were evaluated individually with means and 95% CIs. Data were recorded in a Microsoft Excel spreadsheet, and statistical analysis was performed with R 3.1.1 (R Foundation for Statistical Computing; Vienna, Austria). This study was embedded into a RCT evaluating treatments for T. trichiuris infection, and the sample size was chosen to power this RCT in which our study was nested. We did not conduct a separate sample size analyses and therefore all analyses used this convenience sample and are of exploratory nature rather than conclusive.

Results

Overall, 434 of 594 eligible children completed all baseline and follow-up tests and were included in the final analysis. The mean age of our child cohort was 8.3 years (range 7–14 years) and there were 210 (48.4%) females. Table 1 summarizes children's infection status, stratified by soil-transmitted helminth species. All 434 children were positive for *T. trichiura* at baseline, as this was required for inclusion into the RCT. Across all treatment arms, the prevalence of *T. trichiura* was 70.3% and 71.7% at the 3-week and 18-week follow-up visits, respectively. Overall, *A. lumbricoides* prevalence was 42.2%, 1.6%, and 27.4% at baseline, 3-week, and 18-week posttreatment follow-up visits, respectively. The respective hookworm prevalence was 42.6%, 30.2%, and 30.6%.

Characteristics of soil-transmitted helminth infection	Baseline	3-week treatment follow-up survey	18-week treatment follow-up sur		
Ascaris lumbricoides					
Prevalence (%)	42.2	1.6	27.4		
Prevalence, moderate-heavy infection (%)*	18.0	0.5	5.5		
EPG (95% CI)†	4,319 (3,184; 5,581)	110 (3; 289)	942 (612; 1,318)		
Hookworm					
Prevalence (%)	42.6	30.2	30.6		
Prevalence, moderate-heavy infection (%)*	1.2	0.2	0.0		
EPG (95% CI)†	118 (82; 160)	48 (30; 73)	41 (29; 55)		
Trichuris trichiura					
Prevalence (%)	100	70.3	71.7		
Prevalence, moderate-heavy infection (%)*	30.2	10.1	9.2		
EPG (95% CI)†	1,092 (933; 1,278)	500 (376; 650)	396 (286; 538)		
Strongyloides stercoralis					
Prevalence (%)	2.5	0.2	0.2		

Table 1. Soil-transmitted helminth infection among 434 school-age children on Pemba Island, Tanzania in late 2013/early 2014 at baseline and 3-week and 18-week treatment follow-up surveys.

CI, confidence interval; EPG, eggs per 1 g of feces, computed as arithmetic mean in entire study population, including those without infection.

* Moderate and heavy infection intensities for soil-transmitted helminths were according to guidelines put forth by the World Health Organization (WHO) [3]; *A. lumbricoides*, \geq 5,000 EPG; hookworm, \geq 2,000 EPG; *T. trichiura*, \geq 1,000 EPG

[†] EPG standard errors were computed by bootstrap procedure.

https://doi.org/10.1371/journal.pntd.0007581.t001

There were no history, physical examination, or laboratory (Hb) findings that were associated with infection for *A. lumbricoides*, hookworm, or *S. stercoralis* infection at baseline (Table 2). The composite physical exam finding for anemia, including pallor of the conjunctiva, nail beds, and palmar creases, were associated with lower absolute Hb values (-3.8 g/dl, 95% CI: -6.9, -0.6 g/dl).

Table 2. Relationship between clinical history and physical exam with soil-transmitted helminth infection among
434 school-age children on Pemba Island, Tanzania in a baseline survey conducted in late 2013*.

Description	Ascaris lumbricoides		Hookworm	ı	Strongyloides stercoralis		
	OR (95% CI)	Р	OR (95% CI)	Р	OR (95% CI)	Р	
History							
Abdominal pain	0.22 (0.03, 0.82)	0.04	2.50 (0.85, 8.24)	0.11			
Blood in the stool	2.46 (0.73, 9.50)	0.16	1.13 (0.32, 3.79)	0.85			
Diarrhea			0.22 (0.01, 1.30)	0.16			
Headache	0.68 (0.21, 1.94)	0.48	0.89 (0.29, 2.52)	0.83			
Recent sick days	1.09 (0.53, 2.20)	0.81	1.57 (0.78, 3.20)	0.21			
Wong-Baker Likert scale	1.02 (0.90, 1.16)	0.74	0.94 (0.83, 1.07)	0.35	1.02 (0.68, 1.51)	0.91	
Physical exam							
Signs of anemia	0.80 (0.43, 1.48)	0.49	1.40 (0.76, 2.56)	0.27			
Height (cm)	1.02 (0.99, 1.05)	0.16	1.00 (0.98, 1.02)	0.90	0.99 (0.96, 1.07)	0.78	
Signs of malnutrition	1.25 (0.79, 1.97)	0.34	0.93 (0.59, 1.46)	0.75	1.31 (0.28, 4.64)	0.69	
Laboratory							
Hemoglobin (g/dl)	0.99 (0.98, 1.01)	0.46	0.98 (0.96, 1.00)	0.05	0.98 (0.93, 1.04)	0.53	

CI, confidence interval; OR, odds ratio

*Univariate logistic regressions are presented. Variables without estimates did not include study participants with infections.

https://doi.org/10.1371/journal.pntd.0007581.t002

	Ascaris lumb	ricoides		Hookworm		Strongyloides stercoralis			
	Not infected	Infected	P	Not infected	Infected	P	Not infected	Infected	P
Hb change after treatment (g/dl) ⁱ	+0.01	+0.11	0.19	+0.04	+0.07	0.71	+0.05	+0.04	0.91
	Spearman correlation P Spearman correlation P Spea		Spearman correlation		Р				
Hb change correlation with infection intensity change ⁱⁱ	-0.07		0.17	-0.09		0.07			
	Coefficient		P	Coefficient		Р	Coefficient		P
Wong-Baker Likert scale change, infected <i>versus</i> uninfected (interaction) ⁱⁱⁱ	-0.08		0.37	0.05		0.58	-0.10		0.61

Table 3. Longitudinal relationship of anemia (indirectly assessed by hemoglobin level) and Wong-Baker Likert scale quality of life measure after anthelmintic treatment in a study conducted among school-age children on Pemba Island, Tanzania in late 2013/early 2014*†.

Hb, hemoglobin

* Note: *Trichuris trichiura* was not included in the analysis since only children with *T. trichiura* infection at baseline were included in the trial. [†] Statistical testing for outcomes used: (i) unpaired *t*-test; (ii) Spearman correlation coefficient; and (iii) generalized estimating equation.

https://doi.org/10.1371/journal.pntd.0007581.t003

Table 3 outlines the relationship between soil-transmitted helminth infection intensity with the two key outcomes of faces on Wong-Baker Likert scale and Hb values, both pre- and post-treatment. There was no relation in the face Wong-Baker Likert scale for infected *versus* non-infected individuals, at baseline for *A. lumbricoides*, hookworm, or *S. stercoralis*, or in response to anthelmintic treatment. Similarly, there was no correlation for infection intensity and the Wong-Baker Likert scale for each helminth infection at baseline or in response to treatment.

We measured a borderline association between being infected with hookworm and lower Hb at baseline (-0.2 g/dl Hb, p = 0.05), although the association did not meet significance given correction for multiple comparisons in the analysis. Furthermore, we found a borderline relation between change in Hb after treatment and change in hookworm infection intensity (rho = -0.09, p = 0.07), meaning that reduction in infection intensity after treatment correlated with increases in Hb values. This relation was not present when measuring hookworm infection as a binary variable. We did not find a relation between Hb and infection status for *A*. *lumbricoides* or *S. stercoralis*. Additionally, Hb levels and Wong-Baker Likert scores were comparable between the four treatment arms (Table 4).

Discussion

This study pursued a detailed clinical examination to determine whether soil-transmitted helminth infection cause measurable morbidity in school-age children in a highly endemic area on Pemba Island, Tanzania, and to monitor potential change in morbidity following anthelmintic treatment. We found a weak association between morbidity and soil-transmitted helminth infection, as there was a slight trend toward improving Hb values following the treatment of hookworm infection. This finding corroborates with results from an RCT conducted in 6- to 14-year-old children in Côte d'Ivoire, where anthelmintic treatment slightly improved Hb values [31].

Several non-invasive and non-laboratory-based methods such as questionnaires, have been used to predict and screen for helminth infection, however many of these studies focused specifically on schistosomiasis. Questionnaires have been employed to rapidly screen large cohorts, such as school-age children in endemic settings who are at greatest risk for infection, and demonstrate that simple metrics, such as reported blood in the urine or stool, is associated with *Schistosoma haematobium* and *S. mansoni* infection, respectively [13, 32, 33]. Similarly, simple and rapid clinical tests of morbidity, such as detecting blood in urine with inexpensive urine reagent strips, has demonstrated predictive value for *S. haematobium* diagnoses in

			Albendazole plus oxantel pamoate	Mebendazole	Overall	
Baseline						
Sample size (available data)	108	107	110	110	435	
Mean Hb in g/dl (95% CI)	12.4 (12.3–12.6)	12.6 (12.4–12.8)	12.3 (12.1–12.5)	12.6 (12.4– 12.8)	12.5 (12.4– 12.6)	
Mean Wong-Baker Likert score (95% CI)	2.93 (2.65-3.20)	2.82 (2.52-3.12)	2.83 (2.56–3.09)	2.85 (2.55– 3.16)	2.86 (2.71- 3.00)	
1 st follow-up at 3 weeks posttreatment						
Sample size (available data)	108	105	108	108	429	
Mean Wong-Baker Likert score (95% CI)*	3.19 (2.90-3.47)	3.17 (2.89–3.45)	3.19 (2.89–3.50)	3.16 (2.88– 3.44)	3.18 (3.04– 3.32)	
2 nd follow-up at 18 weeks posttreatment						
Sample size (available data)	105	104	104	107	420	
Mean Hb in g/dl (95% CI)	12.6 (12.3–12.8)	12.7 (12.5–12.9)	12.4 (12.2–12.6)	12.5 (12.3– 12.8)	12.6 (12.5– 12.7)	
Mean Wong-Baker Likert score (95% CI)*	2.72 (2.48–2.96)	2.50 (2.26-2.74)	2.39 (2.13–2.66)	2.45 (2.20– 2.69)	2.51 (2.39– 2.64)	

Table 4. Hemoglobin levels and Wong-Baker Likert scale scores stratified by treatment arms in a study conducted among school-age children on Pemba Island, Tanzania in late 2013/early 2014.

CI, confidence interval; Hb, hemoglobin

https://doi.org/10.1371/journal.pntd.0007581.t004

community settings with high [34, 35] and low *S. haematobium* prevalence [36], and in clinical settings [37].

Despite the long history evaluating rapid, non-invasive, and non-laboratory-based methods for the diagnosis of schistosomiasis, there is a paucity of data evaluating such tools for soil-transmitted helminth infection [38], even though co-infection and polyparasitism is common [39, 40]. Similarly, there is a lack of data using a detailed medical history, coupled with an evidence-based clinical examination to evaluate for soil-transmitted helminth infection using morbidity markers, as done in this study. The benefits of such a detailed clinical approach is that it involves well-trained personnel rather than relying on limited laboratory capacity or expensive equipment [41]. The claim is that an evidence-based clinical approach could detect subtle morbidity (e.g., with questions of general well-being), in addition to evaluating for targeted organ system dysfunction from soil-transmitted helminth infection [42]. In this study, the six-point Wong-Baker Likert scale for general pain and discomfort had no predictive value for soil-transmitted helminth infection or anemia. Although these scales have been validated in children in different socio-cultural settings and are widely used [20, 43, 44], they were not useful in the current setting to distinguish potential reductions in helminth-related morbidity at two time points (3 and 18 weeks) after anthelmintic treatment. However, a detailed clinical approach using a composite of validated physical examination findings was effective at detecting anemia [26, 27]. Such physical exam findings have previously been used to detect anemia in school-age children in African settings [45], and are a simple, rapid, and helpful tool that may prompt public health providers or clinicians to perform a more detailed evaluation for anemia with phlebotomy. While the etiology of anemia in children residing in low- and middle-income countries is multifactorial [46, 47], hookworm infection is known to be a common cause of iron-deficiency anemia [48] and improves with treatment [31, 49]. We detected only a very modest improvement in anemia after anthelmintic therapy at 18 weeks after treatment, and these modest results are likely

due to the relatively low prevalence of moderate and heavy hookworm infection intensity in this cohort, and that we did not provide iron supplementation.

Metrics of morbidity associated with soil-transmitted helminth infection have been detected by a variety of methods, including measuring several anthropometric features such as height, weight, and head circumference in children [50, 51], and physical fitness of both school-age children and adults [52, 53]. Other studies have not demonstrated such associations between soil-transmitted helminth infection and morbidity [54]. Although preventative chemotherapy campaigns aimed at school-age children in high-prevalence settings have shown mixed results in the mitigation of morbidity [8], it is generally believed that periodic deworming is beneficial in these settings, and that some null results are due to a paucity of sensitive morbidity metrics, shorter-term studies, and the dilution of severe morbidity with ongoing preventative chemotherapy efforts [10]. The conflicting results in many of these studies are also reflective of the many challenges to accurately measure soil-transmitted helminth infection-related morbidity in real world settings. This study was conducted in schools with prior soil-transmitted helminthiasis treatment which could have biased our study findings toward demonstrating little benefit of treatment on metrics that measure morbidity.

The current study setting is well suited to evaluate morbidity from soil-transmitted helminthiasis, since other parasitic diseases, such as urogenital schistosomiasis, lymphatic filariasis, and malaria are nearly eliminated [55-57]. Although soil-transmitted helminth re-infection rates in endemic settings as Pemba Island are generally high, there were only relatively few children with moderate-to-heavy hookworm and *A. lumbricoides* infections in the two participating schools, possibly due to frequent treatment within preventive chemotherapy campaigns both for soil-transmitted helminthiasis and lymphatic filariasis, which likely dilutes measurements of morbidity. Other weaknesses of this study include that, by design, all children at baseline were infected with *T. trichiura*, so morbidity from this infection could not be evaluated. This study was integrated into an RCT with a sample size chosen to power that trial, hence our analyses are exploratory in nature. Additionally, only those children who attended school were included in the study, thereby potentially eliminating children who were too unwell to attend school. Finally, the 18 week follow-up period may not have been sufficient time to accurately detect changes in morbidity following anthelmintic treatment.

A detailed clinical evaluation for morbidity in school-age children was not associated with *A. lumbricoides* or *S. stercoralis* infection but demonstrated the utility of the clinical examination to detect anemia, and anemia slightly trended toward resolution after anthelmintic therapy. While many metrics for soil-transmitted helminth infection-related morbidity may have been masked by ongoing preventive chemotherapy efforts, simple physical exam findings for anemia can be useful to highlight individuals that require further evaluation.

Supporting information

S1 STROBE Checklist. (PDF)

Author Contributions

- **Conceptualization:** Isaac I. Bogoch, Benjamin Speich, Said M. Ali, Shaali M. Ame, Jürg Utzinger, Jason R. Andrews, Jennifer Keiser.
- **Data curation:** Isaac I. Bogoch, Benjamin Speich, Nathan C. Lo, David Croll, Said M. Ali, Shaali M. Ame, Jennifer Keiser.

Formal analysis: Isaac I. Bogoch, Benjamin Speich, Nathan C. Lo, Wendelin Moser, Said M. Ali, Shaali M. Ame, Jürg Utzinger, Jason R. Andrews, Jennifer Keiser.

Funding acquisition: Isaac I. Bogoch, Jennifer Keiser.

- **Investigation:** Isaac I. Bogoch, Benjamin Speich, David Croll, Said M. Ali, Shaali M. Ame, Jürg Utzinger, Jason R. Andrews, Jennifer Keiser.
- Methodology: Isaac I. Bogoch, Benjamin Speich, Nathan C. Lo, Wendelin Moser, David Croll, Said M. Ali, Shaali M. Ame, Jürg Utzinger, Jason R. Andrews, Jennifer Keiser.
- **Project administration:** Isaac I. Bogoch, Benjamin Speich, Said M. Ali, Shaali M. Ame, Jennifer Keiser.

Resources: Shaali M. Ame, Jennifer Keiser.

- Supervision: Isaac I. Bogoch, Said M. Ali, Shaali M. Ame, Jürg Utzinger, Jennifer Keiser.
- Validation: Isaac I. Bogoch, Benjamin Speich, Wendelin Moser, Said M. Ali, Shaali M. Ame, Jason R. Andrews, Jennifer Keiser.
- Writing original draft: Isaac I. Bogoch, Benjamin Speich, Said M. Ali, Shaali M. Ame, Jürg Utzinger, Jason R. Andrews, Jennifer Keiser.
- Writing review & editing: Isaac I. Bogoch, Benjamin Speich, Nathan C. Lo, Wendelin Moser, David Croll, Said M. Ali, Shaali M. Ame, Jürg Utzinger, Jason R. Andrews, Jennifer Keiser.

References

- Pullan RL, Smith JL, Jasrasaria R, Brooker SJ. Global numbers of infection and disease burden of soil transmitted helminth infections in 2010. Parasit Vectors. 2014; 7: 37. https://doi.org/10.1186/1756-3305-7-37 PMID: 24447578
- Jourdan PM, Lamberton PHL, Fenwick A, Addiss DG. Soil-transmitted helminth infections. Lancet. 2017; 391 (10117): 252–65. https://doi.org/10.1016/S0140-6736(17)31930-X PMID: 28882382
- **3.** World Health Organization, 2006. Preventive chemotherapy in human helminthiasis: coordinated use of anthelminthic drugs in control interventions: a manual for health professionals and programme managers. Geneva: World Health Organization. Pp. 1–62.
- Lo NC, Bogoch II, Blackburn BG, Raso G, N'Goran EK, Coulibaly JT, et al. Comparison of communitywide, integrated mass drug administration strategies for schistosomiasis and soil-transmitted helminthiasis: a cost-effectiveness modelling study. Lancet Glob Health. 2015; 3 (10): e629–38. <u>https://doi.org/ 10.1016/S2214-109X(15)00047-9 PMID: 26385302</u>
- Jia TW, Melville S, Utzinger J, King CH, Zhou XN. Soil-transmitted helminth reinfection after drug treatment: a systematic review and meta-analysis. PLoS Negl Trop Dis. 2012; 6 (5): e1621. <u>https://doi.org/ 10.1371/journal.pntd.0001621</u> PMID: 22590656
- Lo NC, Addiss DG, Hotez PJ, King CH, Stothard JR, Evans DS, et al. A call to strengthen the global strategy against schistosomiasis and soil-transmitted helminthiasis: the time is now. Lancet Infect Dis. 2017; 17 (2): e64–e9. https://doi.org/10.1016/S1473-3099(16)30535-7 PMID: 27914852
- Taylor-Robinson DC, Maayan N, Soares-Weiser K, Donegan S, Garner P. Deworming drugs for soiltransmitted intestinal worms in children: effects on nutritional indicators, haemoglobin, and school performance. Cochrane Database Syst Rev. 2015; (7): CD000371. https://doi.org/10.1002/14651858. CD000371.pub6 PMID: 26202783
- Welch VA, Ghogomu E, Hossain A, Awasthi S, Bhutta ZA, Cumberbatch C, et al. Mass deworming to improve developmental health and wellbeing of children in low-income and middle-income countries: a systematic review and network meta-analysis. Lancet Glob Health. 2017; 5 (1): e40–e50. https://doi. org/10.1016/S2214-109X(16)30242-X PMID: 27955788
- Bundy D, Peto R. Treatment for intestinal helminth infection. Studies of short term treatment cannot assess long term benefits of regular treatment. BMJ. 2000; 321 (7270): 1225.

- Andrews JR, Bogoch II, Utzinger J. The benefits of mass deworming on health outcomes: new evidence synthesis, the debate persists. Lancet Glob Health. 2017; 5 (1): e4–e5. <u>https://doi.org/10.1016/S2214-109X(16)30333-3</u> PMID: 27955787
- Ezeamama AE, Bustinduy AL, Nkwata AK, Martinez L, Pabalan N, Boivin MJ, et al. Cognitive deficits and educational loss in children with schistosome infection—a systematic review and meta-analysis. PLoS Negl Trop Dis. 2018; 12 (1): e0005524. <u>https://doi.org/10.1371/journal.pntd.0005524</u> PMID: 29329293
- Becker SL, Liwanag HJ, Snyder JS, Akogun O, Belizario V Jr., Freeman MC, et al. Toward the 2020 goal of soil-transmitted helminthiasis control and elimination. PLoS Negl Trop Dis. 2018; 12 (8): e0006606. https://doi.org/10.1371/journal.pntd.0006606 PMID: 30106975
- Lengeler C, Utzinger J, Tanner M. Questionnaires for rapid screening of schistosomiasis in sub-Saharan Africa. Bull World Health Organ. 2002; 80 (3): 235–42. PMID: 11984610
- van der Werf MJ, de Vlas SJ, Brooker S, Looman CWN, Nagelkerke NJD, Habbema JDF, et al. Quantification of clinical morbidity associated with schistosome infection in sub-Saharan Africa. Acta Trop. 2003; 86 (2–3): 125–39. PMID: 12745133
- Speich B, Ali SM, Ame SM, Bogoch II, Alles R, Huwyler J, et al. Efficacy and safety of albendazole plus ivermectin, albendazole plus mebendazole, albendazole plus oxantel pamoate, and mebendazole alone against *Trichuris trichiura* and concomitant soil-transmitted helminth infections: a four-arm, randomised controlled trial. Lancet Infect Dis. 2015; 15 (3): 277–84. <u>https://doi.org/10.1016/S1473-3099(14)</u> 71050-3 PMID: 25589326
- Speich B, Moser W, Ali SM, Ame SM, Albonico M, Hattendorf J, et al. Efficacy and reinfection with soiltransmitted helminths 18-weeks post-treatment with albendazole-ivermectin, albendazole-mebendazole, albendazole-oxantel pamoate and mebendazole. Parasit Vectors. 2016; 9: 123. <u>https://doi.org/ 10.1186/s13071-016-1406-8 PMID: 26935065</u>
- WHO. Guidelines for the use of iron supplements to prevent and treat iron deficiency anemia. Geneva: World Health Organization, 1998. https://www.who.int/nutrition/publications/micronutrients/anaemia_ iron_deficiency/1-57881-020-5/en/ [Date accessed, December 4, 2018].
- Fürst T, Silué KD, Ouattara M, Adiossan LG, N'Goran DN, Yao AJ, et al. Patients routinely report more symptoms to experienced field enumerators than physicians in rural Côte d'Ivoire. Am J Trop Med Hyg. 2013; 89 (3): 592–6. https://doi.org/10.4269/ajtmh.13-0122 PMID: 23878181
- Garra G, Singer AJ, Taira BR, Chohan J, Cardoz H, Chisena E, et al. Validation of the Wong-Baker FACES Pain Rating Scale in pediatric emergency department patients. Acad Emerg Med. 2010; 17 (1): 50–4. https://doi.org/10.1111/j.1553-2712.2009.00620.x PMID: 20003121
- Tomlinson D, von Baeyer CL, Stinson JN, Sung L. A systematic review of faces scales for the self-report of pain intensity in children. Pediatrics. 2010; 126 (5): e1168–98. <u>https://doi.org/10.1542/peds.2010-1609 PMID: 20921070</u>
- 21. Detsky AS, Smalley PS, Chang J. Is this patient malnourished? JAMA. 1994; 271 (1): 54–8. PMID: 8258889
- Williams JW Jr., Simel DL. Does this patient have ascites? How to divine fluid in the abdomen. JAMA. 1992; 267 (19): 2645–8. PMID: 1573754
- 23. Grover SA, Barkun AN, Sackett DL. Does this patient have splenomegaly? JAMA. 1993; 270 (18): 2218–21. PMID: 8411607
- 24. Naylor CD. Physical examination of the liver. JAMA. 1994; 271 (23): 1859–65. PMID: 8196144
- 25. Cook DJ, Simel DL. The rational clinical examination. Does this patient have abnormal central venous pressure? JAMA. 1996; 275 (8): 630–4. PMID: 8594245
- Strobach RS, Anderson SK, Doll DC, Ringenberg QS. The value of the physical examination in the diagnosis of anemia: correlation of the physical findings and the hemoglobin concentration. Arch Int Med. 1988; 148 (4): 831–2.
- Nardone DA, Roth KM, Mazur DJ, McAfee JH. Usefulness of physical examination in detecting the presence or absence of anemia. Arch Int Med. 1990; 150 (1): 201–4.
- Katz N, Chaves A, Pellegrino J. A simple device for quantitative stool thick-smear technique in schistosomiasis mansoni. Rev Inst Med Trop Sao Paulo. 1972; 14 (6): 397–400. PMID: 4675644
- Martin LK, Beaver PC. Evaluation of Kato thick-smear technique for quantitative diagnosis of helminth infections. Am J Trop Med Hyg. 1968; 17 (3): 382–91. https://doi.org/10.4269/ajtmh.1968.17.382
 PMID: 5690644
- Steinmann P, Zhou XN, Du ZW, Jiang JY, Wang LB, Wang XZ, et al. Occurrence of *Strongyloides ster-coralis* in Yunnan Province, China, and comparison of diagnostic methods. PLoS Negl Trop Dis. 2007; 1 (1): e75. https://doi.org/10.1371/journal.pntd.0000075 PMID: 17989788

- Rohner F, Zimmermann MB, Amon RJ, Vounatsou P, Tschannen AB, N'Goran EK, et al. In a randomized controlled trial of iron fortification, anthelmintic treatment, and intermittent preventive treatment of malaria for anemia control in Ivorian children, only anthelmintic treatment shows modest benefit. J Nutr. 2010; 140 (3): 635–41. https://doi.org/10.3945/jn.109.114256 PMID: 20107144
- **32.** Booth M, Mayombana C, Machibya H, Masanja H, Odermatt P, Utzinger J, et al. The use of morbidity questionnaires to identify communities with high prevalences of schistosome or geohelminth infections in Tanzania. Trans R Soc Trop Med Hyg. 1998; 92 (5): 484–90. https://doi.org/10.1016/s0035-9203 (98)90884-7 PMID: 9861358
- Lengeler C, Utzinger J, Tanner M. Screening for schistosomiasis with questionnaires. Trends Parasitol. 2002; 18 (9): 375–7. PMID: 12377245
- Robinson E, Picon D, Sturrock HJ, Sabasio A, Lado M, Kolaczinski J, et al. The performance of haematuria reagent strips for the rapid mapping of urinary schistosomiasis: field experience from Southern Sudan. Trop Med Int Health. 2009; 14 (12): 1484–7. https://doi.org/10.1111/j.1365-3156.2009.02407.x PMID: 19818057
- Ugbomoiko US, Dalumo V, Ariza L, Bezerra FS, Heukelbach J. A simple approach improving the performance of urine reagent strips for rapid diagnosis of urinary schistosomiasis in Nigerian schoolchildren. Mem Inst Oswaldo Cruz. 2009; 104 (3): 456–61. https://doi.org/10.1590/s0074-02762009000300010 PMID: 19547872
- 36. Bogoch II, Andrews JR, Dadzie Ephraim RK, Utzinger J. Simple questionnaire and urine reagent strips compared to microscopy for the diagnosis of *Schistosoma haematobium* in a community in northern Ghana. Trop Med Int Health. 2012; 17 (10): 1217–21. <u>https://doi.org/10.1111/j.1365-3156.2012.03054</u>. x PMID: 22863035
- Ephraim RK, Abongo CK, Sakyi SA, Brenyah RC, Diabor E, Bogoch II. Microhaematuria as a diagnostic marker of *Schistosoma haematobium* in an outpatient clinical setting: results from a cross-sectional study in rural Ghana. Trop Doct. 2015; 45 (3): 194–6. <u>https://doi.org/10.1177/0049475515583793</u> PMID: 25953967
- Fürst T, Ouattara M, Silue KD, N'Goran DN, Adiossan LG, Bogoch II, et al. Scope and limits of an anamnestic questionnaire in a control-induced low-endemicity helminthiasis setting in south-central Côte d'Ivoire. PLoS One. 2014; 8 (6): e64380. <u>https://doi.org/10.1371/journal.pone.0064380</u> PMID: 23755120
- Keiser J, N'Goran EK, Traoré M, Lohourignon KL, Singer BH, Lengeler C, et al. Polyparasitism with Schistosoma mansoni, geohelminths, and intestinal protozoa in rural Côte d'Ivoire. J Parasitol. 2002; 88 (3): 461–6. PMID: 12099412
- Hürlimann E, Yapi RB, Houngbedji CA, Schmidlin T, Kouadio BA, Silué KD, et al. The epidemiology of polyparasitism and implications for morbidity in two rural communities of Côte d'Ivoire. Parasit Vectors. 2014; 7: 81. https://doi.org/10.1186/1756-3305-7-81 PMID: 24568206
- Petti CA, Polage CR, Quinn TC, Ronald AR, Sande MA. Laboratory medicine in Africa: a barrier to effective health care. Clin Infect Dis. 2006; 42 (3): 377–82. https://doi.org/10.1086/499363 PMID: 16392084
- Sackett DL. The rational clinical examination. A primer on the precision and accuracy of the clinical examination. JAMA. 1992; 267 (19): 2638–44. PMID: 1573753
- Gharaibeh M, Abu-Saad H. Cultural validation of pediatric pain assessment tools: Jordanian perspective. J Transcult Nurs. 2002; 13 (1): 12–8. https://doi.org/10.1177/104365960201300103 PMID: 11776010
- Newman CJ, Lolekha R, Limkittikul K, Luangxay K, Chotpitayasunondh T, Chanthavanich P. A comparison of pain scales in Thai children. Arch Dis Child. 2005; 90 (3): 269–70. <u>https://doi.org/10.1136/adc.</u> 2003.044404 PMID: 15723913
- Luby SP, Kazembe PN, Redd SC, Ziba C, Nwanyanwu OC, Hightower AW, et al. Using clinical signs to diagnose anaemia in African children. Bull World Health Organ. 1995; 73 (4): 477–82. PMID: 7554019
- 46. Crawley J. Reducing the burden of anemia in infants and young children in malaria-endemic countries of Africa: from evidence to action. Am J Trop Med Hyg. 2004; 71 (2 Suppl): 25–34. PMID: 15331816
- Righetti AA, Koua AY, Adiossan LG, Glinz D, Hurrell RF, N'Goran EK, et al. Etiology of anemia among infants, school-aged children, and young non-pregnant women in different settings of south-central Côte d'Ivoire. Am J Trop Med Hyg. 2012; 87 (3): 425–34. https://doi.org/10.4269/ajtmh.2012.11-0788 PMID: 22848097
- Farid Z, Miale A Jr. Treatment of hookworm infection in Egypt with bephenium hydroxynaphthoate and the relationship between iron deficiency anemia and intensity of infection. Am J Trop Med Hyg. 1962; 11: 497–505. https://doi.org/10.4269/ajtmh.1962.11.497 PMID: 13891621
- Smith JL, Brooker S. Impact of hookworm infection and deworming on anaemia in non-pregnant populations: a systematic review. Trop Med Int Health. 2010; 15 (7): 776–95. https://doi.org/10.1111/j.1365-3156.2010.02542.x PMID: 20500563

- LaBeaud AD, Nayakwadi Singer M, McKibben M, Mungai P, Muchiri EM, McKibben E, et al. Parasitism in children aged three years and under: relationship between infection and growth in rural coastal Kenya. PLoS Negl Trop Dis. 2015; 9 (5): e0003721. https://doi.org/10.1371/journal.pntd.0003721 PMID: 25996157
- Forrer A, Khieu V, Schär F, Hattendorf J, Marti H, Neumayr A, et al. Strongyloides stercoralis is associated with significant morbidity in rural Cambodia, including stunting in children. PLoS Negl Trop Dis. 2017; 11 (10): e0005685. https://doi.org/10.1371/journal.pntd.0005685 PMID: 29059195
- Müller I, Yap P, Steinmann P, Damons BP, Schindler C, Seelig H, et al. Intestinal parasites, growth and physical fitness of schoolchildren in poor neighbourhoods of Port Elizabeth, South Africa: a cross-sectional survey. Parasit Vectors. 2016; 9 (1): 488. https://doi.org/10.1186/s13071-016-1761-5 PMID: 27595566
- Salmon M, Salmon C, Masoda M, Salumu JM, Bozzi C, Nieburg P, et al. Albendazole treatment improves work capacity in women smallholder farmers infected with hookworm: a double-blind randomized control trial. Am J Trop Med Hyg. 2018; 98 (5): 1419–26. <u>https://doi.org/10.4269/ajtmh.17-0403</u> PMID: 29611504
- Campbell SJ, Nery SV, D'Este CA, Gray DJ, McCarthy JS, Traub RJ, et al. Investigations into the association between soil-transmitted helminth infections, haemoglobin and child development indices in Manufahi District, Timor-Leste. Parasit Vectors. 2017; 10 (1): 192. <u>https://doi.org/10.1186/s13071-017-2084-x</u> PMID: 28424091
- 55. Le Menach A, Tatem AJ, Cohen JM, Hay SI, Randell H, Patil AP, et al. Travel risk, malaria importation and malaria transmission in Zanzibar. Sci Rep. 2011; 1: 93. https://doi.org/10.1038/srep00093 PMID: 22355611
- 56. Knopp S, Person B, Ame SM, Mohammed KA, Ali SM, Khamis IS, et al. Elimination of schistosomiasis transmission in Zanzibar: baseline findings before the onset of a randomized intervention trial. PLoS Negl Trop Dis. 2013; 7 (10): e2474. https://doi.org/10.1371/journal.pntd.0002474 PMID: 24147165
- Rebollo MP, Sambou SM, Thomas B, Biritwum NK, Jaye MC, Kelly-Hope L, et al. Elimination of lymphatic filariasis in the Gambia. PLoS Negl Trop Dis. 2015; 9 (3): e0003642. <u>https://doi.org/10.1371/journal.pntd.0003642</u> PMID: 25785587